**Author's response to reviews**

**Title:** Can father inclusive practice reduce paternal postnatal anxiety? A repeated measures study using the Hospital Anxiety and Depression Scale.

**Authors:**

Jenny Tohotoa (j.tohotoa@curtin.edu.au)
Bruce Maycock (b.maycock@curtin.edu.au)
Yvonne L Hauck (y.hauck@curtin.edu.au)
Satvinder Dhaliwal (s.dhaliwal@curtin.edu.au)
Peter Howat (p.howat@curtin.edu.au)
Sharyn Burns (s.burns@curtin.edu.au)
Colin W Binns (c.binns@curtin.edu.au)

**Version:** 6 **Date:** 19 July 2012

**Author's response to reviews:**

The title page, acknowledgements, conflicting interest and author contributions have been uploaded with the manuscript.